Author's response to reviews

Title: Subannular left ventricular pseudoaneurysm following mitral valve replacement

Authors:

Narayanan Namboodiri (kknamboodiri@yahoo.co.in)
Santosh K Dora (kskdora@hotmail.com)
Bejoy Thomas (bejoy@sctimst.ac.in)
Manoranjan Mishra (mishra@sctimst.ac.in)

Version: 2 Date: 28 February 2008

Author's response to reviews:

To,
The editor,
Journal of cardiothoracic surgery

Dear Sir,

I am resubmitting herewith the manuscript "Subannular left ventricular pseudoaneurysm following mitral valve replacement" after major revision, as suggested by the initial review.

Reviewer's suggestion: For this article to be of benefit to the surgical community, the authors must include a description of the surgical techniques available to correct this anomaly.

A paragraph on the surgical options has been added. The references were modified accordingly.

Thanking you,
Sincerely yours,

Dr. KKN Namboodiri,
Assistant Professor,
Dept. of Cardiology, SCTIMST,